Author's response to reviews

Title: Absence of inferior vena cava in 14-year old boy associated with deep venous thrombosis and positive Mycoplasma pneumoniae serum antibodies- a case report.

Authors:
Boleslaw Kalicki (kalicki@wim.mil.pl)
Monika Sadecka (monika.sdc@gmail.com)
Agata Wawrzyniak (awawrzyniak@wim.mil.pl)
Piotr Kozinski (pkozinski@wim.mil.pl)
Miroslaw Dziekiewicz (dziekiewicz@wim.mil.pl)
Anna Jung (ajung@wim.mil.pl)

Version: 6
Date: 14 March 2015

Author's response to reviews:

The authors updated the Consent section by adding the information that written informed consent for publication of this case report was obtained from parents of the patient.